Author's response to reviews

Title: Cost-effectiveness of the Australian Medical Sheepskin for the prevention of pressure ulcers in somatic nursing home patients: study protocol for a prospective multi-centre randomised controlled trial (ISRCTN17553857)

Authors:

Patriek Mistiaen (p.mistiaen@nivel.nl)
Wilco P Achterberg (wp.achterberg@vumc.nl)
Andre Ament (A.Ament@BEOZ.unimaas.nl)
Ruud Halfens (R.Halfens@zw.unimaas.nl)
Janneke Huizinga (j.huizinga@derm.umcg.nl)
Ken Montgomery (kc-montgomery@netspace.net.au)
Henri Post (hpost@evean.nl)
Anneke L Francke (a.francke@nivel.nl)

Version: 2 Date: 4 January 2008

Author's response to reviews:

Dear editor,

Thank you for the core pre-accept email. We have made the changes as you requested:
- we removed the full stop from the title
- we checked the authors list
- we checked the structure of the abstract and teh manuscript
- we removed website-addresses from the text
- we checked the references, with special attention to the website-references
- we removed the title from figure 1

We hope everything is OK now and ready to publish.

Sincerely

dr. Patriek Mistiaen